Author's response to reviews

Title: Modified technique for transscleral fixation of posterior chamber intraocular lenses

Authors:

Chongde Long (zoclong@21cn.com)
Yantao Wei (weiyantao75@126.com)
Zhaohui Yuan (jorphin@hotmail.com)
Zhiqing Zhang (zhang-zhiqing@sohu.com)
Xiaofeng Lin (linxiaof@mail.sysu.edu.cn)
Bingqian Liu (bingqianliu@gmail.com)

Version: 3 Date: 18 May 2015

Author's response to reviews:

Dear Editor,

We would be grateful for the review of our paper entitled “Modified technique using laminar scleral pockets to cover the long suture ends in transscleral fixation of posterior chamber intraocular lenses”.

We hope this paper will be considered for publication in BMC Ophthalmology. The authors have no proprietary interest in the materials used in this study.

I had full access to all the data in the study and take responsibility for the integrity of the data and the accuracy of the data analysis as well as the decision to submit for publication.

This manuscript has never been previously published in any form by another journal.

Looking forward to hearing from you.

Sincerely Yours,

Bingqian Liu